Author's response to reviews

Title: Adjuvant therapy with minocycline for schizophrenia (The MINOS Trial): Study protocol for a double-blind randomized placebo controlled trial

Authors:

Abebaw Fekadu (abe.wassie@kcl.ac.uk)
Miraf Mesfin (mripta@gmail.com)
Girmay Medhin (gtmedhin@yahoo.com)
Atalay Alem (atalay.alem@gmail.com)
Solomon Teferra (Solomon.teferra@gmail.com)
Tsehaysina Gebre-Eyesus (tsehaysinag@yahoo.com)
Teshale Seboxa (Teshaleseboxa@yahoo.com)
Abraham Assefa (aseffaa@gmail.com)
Jemal Hussien (Jemaldr@gmail.com)
Martha Tibebu (marthatib@yahoo.com)
Christina Borba (cborba@partners.org)
David Hendroson (cdhenderson@partners.org)
Charlotte Hanlon (charlotte.hanlon@kcl.ac.uk)
Teshome Shibre (teshome.kelkile@gmail.com)

Version: 5 Date: 30 October 2013

Author's response to reviews: see over
30 October 2013

To Drs Doug Altman, Curt Furberg, Jeremy Grimshaw and Peter Rothwell

Editors-in-Chief

Trials

RE: MS: 5185336241020649 (Adjuvant therapy with minocycline for schizophrenia (The MINOS Trial): Study protocol for a double-blind randomized placebo controlled trial)

We are again grateful to the reviewer’s comments.

We have now accommodated fully the comments: under Abstract (page 4); under study aims and objectives (page 10); under outcomes (page 20).

Kind regards,

Dr Abebaw Fekadu, on behalf of the authors